ANSWERING REVIEWERS:

There is not much literature on this type of penetrating injury reported elsewhere, hence epidemiological information is limited.

The needed change has been made at the discussion and introduction part.

The loss in shoulder range of motion was due to upper trapezius functional limitation and scarring of tissue at the end of 2 months.

Post discharge rehabilitation has been mentioned in the updated manuscript. Sling was given to the patient postoperatively.